food & nutrition (

ORIGINAL ARTICLE

Dietary patterns in relation to incidence rate of pancreatic cancer – the Norwegian women and cancer cohort study

Eliska Selinger^{1,2}, Charlotta Rylander³ and Guri Skeie^{3*}

¹Department of Epidemiology and Biostatistics, 3rd Faculty of Medicine, Charles University, Prague, Czech Republic; ²Centre for Public Health Promotion, National Institute of Public Health, Prague, Czech Republic; ³Department of Community Medicine, Faculty of Health Sciences, UiT The Arctic University of Norway, Tromsø, Norway

Popular scientific summary

- Prevention of pancreatic cancer is among the priorities of cancer research
- Lifestyle factors are suspected to play a role in the risk of pancreatic cancer
- · Several dietary factors are suspected to influence the risk with conflicting results
- We investigated the relation between whole dietary patterns (in contrast to specific foods) and risk of pancreatic cancer, and the results do not support a role of dietary patterns in the development of pancreatic cancer

Abstract

Despite development in cancer treatment and prevention options during the past few years, cancer of the pancreas remains a diagnosis associated with poor prognosis and limited options for prevention. Diet has proven to be an important risk factor for development of many types of cancer, particularly for cancers of the digestive system. Still, evidence regarding its relation to pancreatic cancer remains ambiguous. To investigate the relationship between diet and pancreatic cancer, an analysis of dietary patterns in participants from the Norwegian Women and Cancer Study (n = 89,156; 305 pancreatic cancer cases) was performed. Cox regression analysis was used for studying possible associations between dietary patterns, derived from principal component analysis, and pancreatic cancer incidence. The four most prominent dietary patterns were identified and described: European pattern, animal food consumers' dietary pattern, traditional Norwegian pattern, and alcohol-abstaining dietary pattern was associated with lower risk of pancreatic cancer in comparison to the lowest tertile (hazard ratios [HR]: 0.66, 95% confidence interval [CI]: 0.49–0.89). After additional adjustment for height and smoking status, no dietary pattern was associated with increased pancreatic cancer risk, nor was there any difference in effect estimates between strata of smokers and non-smokers. The results of our current analysis do not support the role of major dietary patterns in the development of pancreatic cancer.

Keywords: Pancreatic cancer, dietary patterns; women; Norway; prospective cohort

To access the supplementary material, please visit the article landing page

Received: 17 March 2023; Revised: 19 August 2023; Accepted: 23 August 2023; Published: 29 September 2023

Pancreatic cancer is one of the most lethal malignant cancers, responsible for almost half a million deaths per year globally and with a 5-year survival rate of less than 10% (1). Despite these numbers, there is still limited knowledge about different options for primary prevention. According to the World Cancer Research Fund (WCRF) (2), smoking is an established cause of pancreatic cancer. Besides that, there is strong evidence for body fatness being a cause of pancreatic cancer and probable evidence for adult height. However, only limited evidence currently exists for a relationship between specific dietary factors such as red and processed meat, foods containing saturated fatty acids, alcoholic drinks and foods and beverages containing fructose, and development of pancreatic cancer.

Research on nutritional factors involved in cancer development is complicated by several limitations. First, the relatively low incidence of many cancers and long latency period require large cohort studies with long follow-up to gain sufficient power to detect important differences. Moreover, the potential effect of nutrition is confounded by a range of non-dietary factors, like other lifestyle characteristics (e.g. physical activity) or socio-economic status as well as by intake of other dietary groups, challenging the interpretation of any findings. Single nutrients or single food usually have a limited impact on health outcomes. Moreover, when assessing overall health effects, it is essential to consider not only the specific food under investigation but also the potential effect of different substitutes if consumption changes. The interpretation of single-food studies is therefore limited as it may overlook the complexity of different dietary interactions existing in a real world. This has led to heated discussions about the reliability of such an approach, especially due to many discordances between findings of cohort studies and subsequent trials (3). In an attempt to at least partially compensate for these limitations associated especially with the single food approach, many advocate for a shift towards investigation of whole dietary patterns, which tend to bring more robust results across different types of studies, arguing to preserve single-nutrient analysis only as additional evidence (4-6).

Several methods of investigation of dietary patterns and their impact on health outcomes have been described (7, 8). Beside investigator-driven methods, using pre-specified dietary scores and pre-defined dietary patterns, datadriven methods have come to use. The basis for them is the utilization of exploratory statistical methods, like clustering methods, to identify dietary patterns naturally occurring in the population without any predetermined set of criteria. Characteristics related to the identified patterns can then be further investigated, including testing of their association with disease occurrence. One of the exploratory methods used for the identification of such habitual dietary patterns is the principal component analysis (PCA) (3, 9–11).

To investigate the potential relationship between diet and the incidence of pancreatic cancer, we decided to utilize data from a cohort of Norwegian women. The main aims of the present study are to identify dietary patterns adopted by the Norwegian female population using PCA and to analyze their relationship with the risk of pancreatic cancer.

Methods

Study design, setting, and available data

The Norwegian Women and Cancer (NOWAC) cohort study is a cohort recruited between the years 1991 and 2005. Women were randomly selected from the National registry and invited to participate in the study. A detailed profile of the cohort has been published elsewhere (12). Most of the recruited women completed a detailed validated food frequency questionnaire (FFQ) and self-reported important health information, for example, height, weight, use of hormonal contraceptives, hormonal replacement therapy, smoking, as well as socio-economic data like educational level and number of children (13-15). The FFO used for recording of dietary information was validated in several methodological studies, including a biomarker study, a comparison with repeated 24 h dietary recalls, and used in a reproducibility study. When compared to 24 h dietary recalls, it was concluded that the FFQ demonstrated a commendable capacity to rank participants concerning frequently consumed foods (14-16). The data were linked with the Cancer Registry of Norway. The NOWAC study is conducted by a team of researchers mainly at UiT the Arctic University of Norway. This study was performed in line with the principles of the Declaration of Helsinki. Approval was granted by the Regional Committee for Medical Research Ethics in Northern Norway (P REK NORD 141/2008) and the Norwegian Data Inspectorate. Informed consent was obtained from all participants. The study results are reported in line with STROBE-nut guideline (17).

Participants

A sample of 95,937 women was selected from study participants recruited in several waves between 1995 and 2005. Women reporting extreme energy intake in the food-frequency questionnaire (under 2,500 kJ or over 15,000 kJ) were excluded to decrease the risk of unrealistic dietary reporting distorting the results of the dietary patterns analysis (n = 1,053). Moreover, due to the planned survival analysis, women with missing information on smoking status (n = 1,224), height (n = 969), or women with previous cancer diagnoses were excluded before the analysis of dietary patterns (n = 3,835), with some of the participants missing information on more than one of these variables. Therefore, altogether, 6,781 persons were excluded from the analyses. The resulting sample consisted of 89,156 cancer-free women. In the final sample, 305 new cases of pancreatic cancer were recorded during the follow-up period. Details of the selection process are presented in Fig. 1.

Principal component analysis

For the PCA, 73 to 90 original variables were retrieved from the FFQ and pre-grouped into 39 food variables (see Supplementary Table 1 for the detailed list of specific foods included in the grouped food variables). It is known that these decisions play a crucial role in the estimations of disease-exposure relationship, with research indicating that more detailed information on food groups intake results in enhanced precision of disease risk assessments (18). Due to this, decision on grouping was based not only on the nutritional value of a given food, but also on its known association with health and usage in Norwegian cuisine. Variables containing data on sweet beverages and nuts consumption



Fig. 1. Flowchart of the selection process. Participants with unrealistic energy intakes or missing information on confounders/effect modifiers were excluded from the analysis. *Some participants had missing values for several variables, so the total number of excluded participants (6,781) is lower than the sum of the missing values.

were excluded from the analysis, as not all participants were asked about their intake in the FFO used in their wave of recruitment. After the re-grouping and exclusion, Bartlett test of sphericity and Kaiser-Mayer-Olkin tests were performed to confirm the suitability of data for the analysis. The Bartlett's test of sphericity was significant at an alpha level of 0.05 (P < 0.001), with overall Keiser-Meyer-Olkin value of 0.74. PCA was then applied to identify dietary patterns present in the obtained data. The food group intake values were standardized to reduce the influence of food intakes with high variance in the PCA and varimax rotation was performed on the loadings of PCA. Visual inspection of eigenvalues and percentage of variance explained were used as a measure for identification of the most prominent principal components (PCs) and their main dietary components were described in detail (main variables are defined as the 15 variables with the highest absolute value of eigenvalues). Full scree plots obtained from PCA analysis, including scree plots of variable loadings for selected components, can be found in Supplementary Figs. 1 and 2. Individual scores for selected PCs were retrieved (Supplementary Table 2), divided into tertiles, and associated background characteristics were assessed: age (in years), body mass index (BMI) (in kg/m² and divided into two categories: <25 kg/m² and BMI \ge 25 kg/m²), height (in cm), number of children (categorized as no children, 1-2 children, more than two children), education level (years of schooling, categorized as 0-9 years, 10-12 years, more than 12 years)

and smoking status (current smoker, previous smoker, nonsmoker). The characteristics were described as mean and standard deviations or proportions, in case of categorical variables. All work was done using R software version 4.2.2, the FactoMineR (19) and factoextra (20) packages.

Survival analysis

Cox regression models were built to describe the relationships between the identified dietary patterns, used as an independent exposure, and incidence of pancreatic cancer as outcome. Incident cases of pancreatic cancer were defined as those given code C25 from the International Classification of Diseases version 10 (ICD-10). For the purpose of the survival analysis, attained age in years was used as time scale. The end of follow-up was defined as the age at the end of the study period (31.12.2016), age at emigration, age at death, or cancer diagnosis, whichever came first.

Three models were constructed, one with mutual adjustment for identified dietary patterns, and two multivariable-adjusted models, one adjusting for height and smoking and another one with additional adjustment for BMI. Based on previous research describing their strong relation to the risk of pancreatic cancer, smoking status (non-smokers, ex-smokers, smokers) and height were used as additional independent variables in the multivariable model 2. Moreover, analyses stratified on smoking status were performed due to the possibility of effect modification of the dietary risk. Smoking status was divided into two strata based on baseline reported smoking status (current non-smokers, current smokers). BMI and energy intake were considered to be mediators of the dietary effect and therefore was not included in the main models. However, it is important to take into account also the possibility that the relationship between diet and BMI can be more complex. Diet itself can influence the body weight, but BMI can also influence the reported dietary intake. Also, there was a relationship with dietary intake as well as tendency for the association with the outcome of interest in our dataset (mean BMI in cases = 25.2 vs. non-cases 24.7; P = 0.05). Therefore, an additional model with BMI added as a confounder was constructed to explore the direct association between any of the dietary patterns not mediated through the effect of body weight. Results of the analyses are reported as hazard ratios (HR), together with corresponding 95% confidence intervals (CI). Statistical analysis was undertaken using R software version 4.2.2, Survival package (21).

Results

Characteristics of the sample

After the exclusions, 89,156 women were included in the analyses contributing with 1,424,465.9 person-years of follow-up. The median follow-up time of participants was 15.9 years (min: 0.03 years, max: 20.58 years). Mean age at the time of enrolment was 51.6 years. Detailed characteristics of the final sample are presented in Table 1.

Dietary patterns

After the PCA was performed and scree plot examined, four main dietary patterns were identified: European pattern, animal food consumers' dietary pattern, traditional Norwegian pattern, and alcohol-abstaining dietary pattern. A detailed description of the loadings of each variable associated with the identified dietary patterns after varimax rotation can be found in Fig. 2, Supplementary Table 3 and Supplementary Fig. 3. Description of relevant background characteristics of participants per pattern is presented in Supplementary Table 2.

PC1: European diet

This dietary pattern was characterized by high consumption of vegetables (particularly salad vegetables and other vegetables), fruit, pasta, or rice, but also pizzas, waffles, and sweets. As protein, chicken, and yoghurt were prominent. Potatoes and fish spreads were consumed in lower amounts than in other patterns. Participants with high scores for this pattern tended to be relatively younger, more educated, and not have children. Also, the highest tertile had a lower proportion of current smokers compared to the lowest tertile.

PC2: Animal food consumers

This dietary pattern was characterized by high consumption of meat, including processed and red meat

Baseline characteristics	Mean	SD
Age at enrolment (years)	51.6	6.3
Height (cm)	166.2	5.7
Weight (kg)	68.3	11.6
BMI (kg/m²)	24.7	4.0
	n	Percentage, %
Smoking status		
Non-smoker	33,227	35
Previous smoker	32,590	35
Smoker	27,924	30
BMI category		
BMI < 25 kg/m ²	55,475	60
BMI \geq 25 kg/m ²	36,367	40
Number of years in school		
0–9 years	23,133	26
10–12 years	30,306	34
More than 12 years	35,204	40
Number of children		
No children	8,100	9
I-2 children	49,067	52
More than two children	36,574	39

*Total number of participants may differ between the variables due to various missing values.

and meat spreads, eggs, and animal fat. Reported drinking of hard liquors and beer is also apparent. Women in the highest tertile of the score for this pattern were younger compared to those in the lowest tertile, less educated, tended to have more children and were current smokers.

PC3: Traditional Norwegian diet

The traditional Norwegian dietary pattern was characterized by a high intake of all kinds of fish and fish products, including cod liver oil, fruit, and vegetables (especially root vegetables) and potatoes. At the same time, participants with high scores for this pattern tended to eat less salad vegetables, sweets, and pizzas. Women in the highest tertile for this pattern tended to be older on average, less educated and non-smokers compared to women in the lowest tertile.

PC4: Abstainers

The most prominent characteristic of the fourth dietary pattern was the low consumption of all kinds of alcohols (liquors, wine, beer), accompanied by higher consumption of wholemeal bread, porridge, cakes, jam, and sweets. Participants in the highest tertile were more educated, more likely to be non-smokers and a higher proportion of them had more than two children compared to women in the lowest tertile.

Risk of pancreatic cancer

Being in the highest tertile for the abstaining dietary pattern was inversely associated with pancreatic cancer risk compared to the lowest tertile (HR 0.66, 95% CI: 0.49–0.89) in the model with mutual adjustment for all four dietary patterns. No clear relationship was apparent with the other dietary patterns. After adjustment for height and smoking status, we observed no substantial association between any of the dietary patterns and incidence of pancreatic cancer. Additionally, a model adjusted also for the BMI status (normal weight or overweight and obese) was prepared to investigate the direct association of dietary patterns on pancreatic cancer without confounding by body weight. There were no notable changes in the estimates after this additional adjustment. Detailed results for unadjusted and adjusted models are reported in Table 2.

Stratification

While smoking status by itself was associated with the incidence of pancreatic cancer, no dietary pattern was associated with pancreatic cancer risk in strata of current smokers or non-smokers, adjusted for height. The full results of the stratified analyses are shown in Table 3.



Fig. 2. Principal component analysis: Main components loadings for four most prominent dietary patterns identified.

		n	n of cases	Model I	Model 2	Model 3
				HR (95% CI)	HR (95% CI)	HR (95% CI)
PC1: European	l st tertile	29,719	134	Ref.	Ref.	Ref.
	2nd tertile	29,718	103	1.35 (1.04 – 1.76)	1.42 (1.09 – 1.84)	1.43 (1.10 – 1.86)
	3rd tertile	29,719	64	0.83 (0.60 - 1.16)	0.92 (0.66 - 1.0)	0.92 (0.66 - 1.29)
PC2:Animal	l st tertile	29,719	112	Ref.	Ref.	Ref.
food consumers	2nd tertile	29,718	97	1.09 (0.83 - 1.44)	1.07 (0.81 – 1.41)	1.04 (0.79 – 1.38)
	3rd tertile	29,719	92	1.06 (0.80 - 1.41)	0.98 (0.74 - 1.31)	0.97 (0.73 – 1.29)
PC3:Traditional	l st tertile	29,719	75	Ref.	Ref.	Ref.
Norwegian	2nd tertile	29,718	92	0.96 (0.71 – 1.30)	0.99 (0.73 – 1.34)	0.95 (0.69 - 1.29)
	3rd tertile	29,719	134	0.97 (0.72 -1.32)	1.03 (0.77 – 1.40)	1.02 (0.76 – 1.38)
PC4: Abstainers	l st tertile	29,719	123	Ref.	Ref.	Ref.
	2nd tertile	29,718	109	0.87 (0.67 - 1.12)	1.02 (0.78 - 1.33)	1.03 (0.79 – 1.34)
	3rd tertile	29,719	69	0.66 (0.49 - 0.89)	0.86 (0.63 - 1.16)	0.85 (0.62 - 1.16)

Table 2. Association between dietary patterns and risk of pancreatic cancer in the Norwegian Women and Cancer study. Tertiles of individual scores for dietary patterns compared.

Model I: Mutual adjustment of the four dietary patterns only.

Model 2: Model 1 + adjustment for height and smoking.

Model 3: Model 2 + BMI group.

Discussion

We revealed four main dietary patterns among Norwegian women: a European pattern, characterized by high intake of fruit, vegetables, pasta and rice, lean meat but also sweets, an animal food consumers' pattern with a high intake of all kinds of animal-based food, except for fish, a traditional Norwegian pattern based mainly on the consumption of fish and fish products together with potatoes and root vegetables, and an alcohol-abstaining dietary pattern, characterized by the low intake of all kinds of alcoholic drinks. Despite some evidence for a relation between the alcohol-abstaining dietary pattern and the risk of pancreatic cancer in the analysis with mutual adjustment for all four dietary patterns, no dietary

		Current smokers	Current non-smokers	
Total <i>n</i> of participants:		26,717	62,439	
Total <i>n</i> of events:		143	158	
Multivariable model*		HR (95% CI)	HR (95% CI)	
PC1: European	l st tertile	Ref.	Ref.	
	2nd tertile	1.91 (1.32 – 2.76)	1.04 (0.71 – 1.52)	
	3rd tertile	0.77 (0.44 – 1.32)	I.00 (0.65 – I.53)	
PC2:Animal	l st tertile	Ref.	Ref.	
food consumers	2nd tertile	I.27 (0.83 – I.94)	0.93 (0.65 – 1.35)	
	3rd tertile	I.22 (0.80 – I.86)	0.83 (0.56 - 1.23)	
PC3:Traditional	l st tertile	Ref.	Ref.	
Norwegian	2nd tertile	0.97 (0.64 - 1.46)	1.05 (0.66 – 1.67)	
	3rd tertile	0.83 (0.54 - 1.28)	I.27 (0.82 – I.96)	
PC4: Abstainers	l st tertile	Ref.	Ref.	
	2nd tertile	I.08 (0.75 – I.58)	0.91 (0.63 - 1.32)	
	3rd tertile	I.07 (0.68 – I.67)	0.70 (0.47 - 1.05)	

Table 3. Association between dietary patterns and risk of pancreatic cancer in the Norwegian Women and Cancer study, stratified by smoking status. Tertiles of individual scores for dietary patterns compared, adjusted for height.

*Results from Cox regression, with mutually adjusted dietary patterns and adjustment for height.

pattern was found to have any apparent association with the risk of pancreatic cancer development after adjustment for known confounders.

As there is a possibility of effect modification by smoking status, or in other words diet having different associations with pancreatic cancer in smokers versus non-smokers, a stratified analysis was performed. No strong evidence for any relationship was found for any of the strata or dietary patterns. Neither did our estimates change much when the main model was additionally adjusted for BMI.

Several previous studies have attempted to describe the possible relation between food intake, diet, or lifestyle and pancreatic cancer. No relationship was found with intake of vegetables, fruit, or fiber (22). Another analysis of the European Prospective Investigation into Cancer and Nutrition (EPIC) cohort data observed a relationship between a healthy lifestyle index and pancreas cancer. High score on a healthy lifestyle index based on a combination of smoking, alcohol intake, dietary exposure, physical activity, and waist-to-hip was associated with a lower hazard of pancreas cancer. After the removal of smoking, Population attributable fraction (PAF) for pancreas cancer and lifestyle was estimated as 14% (95% CI: 6-21) (23). Moreover, two previously published systematic reviews and meta-analyses identified an inverse association of risk of pancreatic cancer and patterns with high intake of fiber, vitamins and vegetables or light-drinking of alcohol, while reporting positive associations between pancreatic cancer risk and Western dietary patterns (24, 25). Regarding the relationship between pancreatic cancer risk and dietary patterns, a recent umbrella review published in 2022 recognized an association between healthy plant-based diet and lower risk of pancreatic cancer. Conversely, high intake of red meat was associated with increased risk of pancreatic cancer. However, the general quality of evidence was low to critically low (26). Similar results were obtained in a multi case-control study published shortly after the umbrella review. Its aim was to evaluate the relationship between previously defined pro-vegetarian dietary patterns and risk of selected gastrointestinal cancers. In this study, comparing the relative risk ratios (RR) of the highest and the lowest quintiles for healthy pro-vegetarian pattern found a higher adherence to healthy pro-vegetarian pattern to be associated with a lower risk of pancreas cancer (RRR: 0.74, 95% CI: 0.59-0.92) (27). Those findings are in line with another meta-analysis concluding that plantbased diet is associated with lower risk of digestive system neoplasms, especially with pancreatic cancer (adjusted RR 0.71, 95% CI: 0.59-0.86) (28). In our study, we did not identify one specific pattern as plant-based, as different plant-based food groups loaded more strongly for different patterns, except the animal food consumption pattern.

Our results therefore differ from the previously reported studies as they do not confirm strong association between any identified dietary pattern and pancreatic cancer. Possible reasons for the differing conclusion can be different background risk in the populations involved as well as differences in dietary assessment methods or the methods used to identify the score for the dietary pattern with some studies using pre-specified criteria while others relying on posteriori methods, for example, cluster, factor or PCA. Differences in definitions and statistical methods used can lead to miscellaneous food groups combined in the patterns and can be the underlying reasons for the heterogeneous conclusions reported in available research. On the other hand, the most up-to-date report by WCRF still mentions only two confirmed causes of pancreatic cancer with strong evidence (having overweight or obesity, and being tall), and several other causes with some evidence (red meat, processed meat, smoking, and alcoholic drinks) (2). Our results add to the evidence of no clear association between dietary patterns including dietary patterns characterized by relatively high intakes of processed and red meat. In our results, there was tendency of lower risk associated for the dietary pattern with low alcohol consumption, but after adjustment for other confounders, even that relationship disappeared.

There are several limitations in our analysis. First, dietary data as well as other relevant information (smoking status and height) were self-reported by the participants. This creates potential for measurement bias which could lower our confidence in the obtained results. To limit this effect, only validated FFQs were used to collect the information, and participants with unrealistically reported intakes (defined as energy intakes under 2,500 kJ or over 15,000 kJ) were excluded from the analysis. Even though the number of cases in the total sample should be sufficient for the main analysis (n = 305), in the stratified analysis the number of cases per group was lower, reducing the overall power of the analysis. Therefore, the results of the stratified analyses should be interpreted with caution. To reduce confounding, multivariable models were constructed accounting for confounders selected based on available evidence. However, there could be other potential risk factors for pancreatic cancer that we could not adjust for, for example, family history. Another limitation is the inability to account for a family history or current diagnosis of diabetes, despite diabetes being a known risk factor for the development of pancreatic cancer (29). Due to this, residual confounding cannot be excluded.

Conclusion

We observed no association between dietary patterns and risk of pancreatic cancer, nor was there any difference in associations between smokers and non-smokers. The results of our current analysis do not support a role of major dietary patterns in the development of pancreatic cancer.

Conflict of interest and funding

The publication charges for this article have been funded by a grant from the publication fund of UiT the Arctic University of Norway, and the Healthy Choices project (ref. 289440). The work performed was funded by the Erasmus+ Programme. The authors declare no potential conflicts of interest.

Acknowledgements

We thank all the participants who contributed information to the NOWAC study. The study has used data from the Cancer Registry of Norway. The interpretation and reporting of these data are the sole responsibility of the authors, and no endorsement by the Cancer Registry of Norway is intended nor should be inferred.

References

- Rawla P, Sunkara T, Gaduputi V. Epidemiology of pancreatic cancer: global trends, etiology and risk factors. World J Oncol 2019; 10(1): 10–27. doi: 10.14740%2Fwjon1166
- World Cancer Research Fund/American Institute for Cancer Research. Continuous Update Project Expert Report 2018. Diet, nutrition, physical activity, and pancreatic cancer. [cited: 20 September 2023]. Available from: https://www.wcrf.org/sites/ default/files/Pancreatic-cancer-report.pdf
- Mozaffarian D, Rosenberg I, Uauy R. History of modern nutrition science-implications for current research, dietary guidelines, and food policy. BMJ 2018; 361: k2392. doi: 10.1136/bmj.k2392

- Tapsell LC, Neale EP, Satija A, Hu FB. Foods, nutrients, and dietary patterns: interconnections and implications for dietary guidelines. Adv Nutr 2016; 7(3): 445–54. doi: 10.3945/an.115.011718
- McGuire S. Scientific Report of the 2015 Dietary Guidelines Advisory Committee. Washington, DC: US Departments of Agriculture and Health and Human Services, 2015. Adv Nutr 2016; 7: 202–4.
- Cespedes EM, Hu FB. Dietary patterns: from nutritional epidemiologic analysis to national guidelines. Am J Clin Nutr 2015; 101(5): 899–900. doi: 10.3945/ajcn.115.110213
- Zhao J, Li Z, Gao Q, et al. A review of statistical methods for dietary pattern analysis. Nutr J. 2021; 20(1): 37. doi: 10.1186%2Fs12937-021-00692-7
- DiBello JR, Kraft P, McGarvey ST, Goldberg R, Campos H, Baylin A. Comparison of 3 methods for identifying dietary patterns associated with risk of disease. Am J Epidemiol 2008; 168(12): 1433–43. doi: 10.1093%2Faje%2Fkwn274
- Hu FB. Dietary pattern analysis: a new direction in nutritional epidemiology. Curr Opin Lipidol 2002 Feb; 13(1): 3–9. doi: 10.1097/00041433-200202000-00002
- Hu FB, Rimm E, Smith-Warner SA, Feskanich D, Stampfer MJ, Ascherio A, et al. Reproducibility and validity of dietary patterns assessed with a food-frequency questionnaire. Am J Clin Nutr 1999; 69(2): 243–9. doi: 10.1093/ajcn/69.2.243
- Hu FB, Rimm EB, Stampfer MJ, Ascherio A, Spiegelman D, Willett WC. Prospective study of major dietary patterns and risk of coronary heart disease in men. Am J Clin Nutr 2000; 72(4): 912–21. doi: 10.1093/ajcn/72.4.912
- Lund E, Dumeaux V, Braaten T, Hjartaker A, Engeset D, Skeie G, et al. Cohort profile: the Norwegian women and cancer study–NOWAC–Kvinner og kreft. Int J Epidemiol 2007; 37(1): 36–41. doi: 10.1093%2Fije%2Fdym137
- Lund E, Kumle M, Braaten T, Hjartåker A, Bakken K, Eggen E, et al. External validity in a population-based national prospective study The Norwegian Women and Cancer Study (NOWAC). Cancer Causes Control 2003; 14(10): 1001–8. doi: 10.1023/b:caco.0000007982.18311.2e
- 14. Parr CL, Veierød MB, Laake P, Lund E, Hjartåker A. Testretest reproducibility of a food frequency questionnaire (FFQ) and estimated effects on disease risk in the Norwegian Women and Cancer Study (NOWAC). Nutr J 2006 Jan 31; 5: 4. doi: 10.1186/1475-2891-5-4
- Hjartåker A, Andersen LF, Lund E. Comparison of diet measures from a food-frequency questionnaire with measures from repeated 24-hour dietary recalls. The Norwegian Women and Cancer Study. Public Health Nutr 2007 Oct; 10(10): 1094–103. doi: 10.1017/S1368980007702872
- Hjartåker A, Lund E, Bjerve KS. Serum phospholipid fatty acid composition and habitual intake of marine foods registered by a semi-quantitative food frequency questionnaire. Eur J Clin Nutr 1997 Nov; 51(11): 736–42. doi: 10.1038/ sj.ejcn.1600475
- 17. Lachat C, Hawwash D, Ocké MC, Berg C, Forsum E, Hörnell A, et al. Strengthening the Reporting of Observational Studies in Epidemiology-Nutritional Epidemiology (STROBE-nut): an extension of the STROBE statement. PLoS Med 2016; 13(6): e1002036. doi: 10.1371%2Fjournal.pmed.1002036
- McCann SE, Marshall JR, Brasure JR, Graham S, Freudenheim JL. Analysis of patterns of food intake in nutritional epidemiology: food classification in principal components analysis and the subsequent impact on estimates for endometrial cancer. Public Health Nutr 2001; 4(5): 989–97. doi: 10.1079/ phn2001168

- Lê S, Josse J, Husson F. FactoMineR: An R Package for Multivariate Analysis. J. Stat. Soft. 2008; 25(1): 1–18. doi: 10.18637%2Fjss.v025.i01
- Kassambara A, Mundt F. Factoextra: Extract and Visualize the Results of Multivariate Data Analyses. R Package Version 1.0.7. 2020. [cited: 20 September 2023]. Available from: https://CRAN.Rproject.org/package=factoextra
- Therneau T, Lumley T. Package Survival: A Package for Survival Analysis in R. R Package. Version 2, 38. 2015. [cited: 20 September 2023]. Available from: https://CRAN.R-project.org/ package=survival
- Bradbury KE, Appleby PN, Key TJ. Fruit, vegetable, and fiber intake in relation to cancer risk: findings from the European Prospective Investigation into Cancer and Nutrition (EPIC). Am J Clin Nutr 2014; 100 Suppl 1: 394S–8S. doi: 10.3945/ ajcn.113.071357
- Naudin S, Viallon V, Hashim D, Freisling H, Jenab M, Weiderpass E, et al. Healthy lifestyle and the risk of pancreatic cancer in the EPIC study. Eur J Epidemiol 2020; 35(10): 975–86. doi: 10.1007/s10654-019-00559-6
- Zheng J, Guinter MA, Merchant AT, Wirth MD, Zhang J, Stolzenberg-Solomon RZ, et al. Dietary patterns and risk of pancreatic cancer: a systematic review. Nutr Rev 2017 Nov 1; 75(11): 883–908. doi: 10.1093/nutrit/nux0380
- Lu PY, Shu L, Shen SS, Chen XJ, Zhang XY. Dietary patterns and pancreatic cancer risk: a meta-analysis. Nutrients 2017; 9(1): 38. doi: 10.3390/nu9010038

- Gianfredi V, Ferrara P, Dinu M, Nardi M, Nucci D. Diets, dietary patterns, single foods and pancreatic cancer risk: an umbrella review of meta-analyses. Int J Environ Res Public Health 2022; 19(22): 14787. doi: 10.3390/ijerph192214787
- 27. Oncina-Cánovas A, González-Palacios S, Notario-Barandiaran L, Torres-Collado L, Signes-Pastor A, de-Madaria E, et al. Adherence to pro-vegetarian food patterns and risk of oesophagus, stomach, and pancreas cancers: a multi case-control study (The PANESOES Study). Nutrients 2022; 14(24): 5288. doi: 10.3390/nu14245288
- Zhao Y, Zhan J, Wang Y, Wang D. The relationship between plant-based diet and risk of digestive system cancers: a meta-analysis based on 3,059,009 subjects. Front Public Health 2022; 10: 892153. doi: 10.3389/fpubh.2022.892153
- Roy A, Sahoo J, Kamalanathan S, Naik D, Mohan P, Kalayarasan R. Diabetes and pancreatic cancer: exploring the two-way traffic. World J Gastroenterol 2021; 27(30): 4939–62. doi: 10.3748/wjg.v27.i30.4939

*Guri Skeie

Department of Community Medicine, Faculty of Health Sciences, UiT The Arctic University of Norway, Postbox 6050 Langnes, 9037 Tromsø, Norway. Email: guri.skeie@uit.no